Author's response to reviews

Title: Development of the Tilburg Pregnancy Distress Scale: the TPDS

Authors:

Victor J.M. Pop (V.J.M.Pop@uvt.nl)
Antoinette M. Pommer (A.M.Pommer@uvt.nl)
Monica Pop-Purceleanu (M.Pop-Purceleanu@psy.umcn.nl)
Hennie A.A. Wijnen (HAAWijnen@onsbrabantnet.nl)
Veerle Bergink (V.Bergink@erasmusmc.nl)
Frans Pouwer (F.Pouwer@uvt.nl)

Version: 4 Date: 20 October 2011

Author's response to reviews: see over
Dear Editor,

Please find enclosed the final version of our paper: Development of the Tilburg Pregnancy Distress Scale: the TPDS. All proposed changes to the format of the manuscript so that the published article conforms to the journals style have been adopted.

**Major revisions**

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system.

**Minor revisions**

Affiliations: Each affiliation should be written in full, in the following format: department, institute, (street), city, (postal code), country.

Figures: it is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimize white space around the image.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We hope that our revised version is suitable for publication in your journal.

Sincerely yours,

On behalf of the authors.